\begin{table}%T4  
\caption{\label{tabNewVar}New variable~stars.}
%\begin{center}
{
\begin{tabular}{lllcrllllll} 
\hline\hline
Sonneberg&Hip or &Spec.&    $N$  & Mean &Drift&\multicolumn{2}{l}{Periodic Var.} & \multicolumn{2}{l}{ Erratic Variable} & Remarks \\
variable&GSC No. & type&         & ~mag  &  ampl.  &     $P$(d) & ampl. &     Time scale &  ampl. &  \\
designation&        &          &     &  $B$     & (mag)         &     & (mag)   &
  (d)  & (mag) &  \\ \hline
S10949&Hip 25972 &A0&   478&    8.547& ~$ -0.15$&        &       &       30$-$8000 &  0.35    &  \\
S10950&Hip 26162 &B8e&  467&    8.570& ~$ -0.22$&        &       &     1000$-$4000 &  0.5&\\
S10951&Hip 26669 &A0&   497&    9.027&          & 8000:  & 0.2   &       30$-$300  &  0.15&     see Fig.~\ref{S10951}\\
S10952&Hip 27699 &B5e&  477&    9.010&     +0.26& 4700   & 0.25  &               &      &    see Fig.~\ref{S10952}\\
S10953&0708.0904 &M0&   473&   11.158&          &        &       &
&  & =NSV 2001,$(1)$,~$\sigma = 0.186$~mag \\
S10954&0709.0746 &A5&   425&    9.778&     +0.21&        &       &               &&\\
S10955&0714.0246 &A0&   481&    9.281&          & 1.58625& 0.2   &               &      & ~$(2)$, see Fig.~\ref{S10955} \\
S10956&0714.1144 &K0&   476&    9.578&          & 11000: & 0.2   &               &      & "wave'' see Fig.~\ref{S10956}\\
S10957&1873.0775 &M &   468&   11.026&          &        &       &      4000$-$6000&  0.3  &=NSV 2444, Fig.~\ref{S10957}\\
S10958&1875.2587 &A2&   474&    8.182&     +0.18&        &       &               &        &        =NSV 2670\\
S10959&2403.0702 &A5m&  461&   10.265&     +0.17&  8000: & 0.15  &               &         &       see Fig.~\ref{S10959}\\
S10960&2405.0168 &A0&   461&   10.764&     +0.09&        &       &               &&\\
S10961&2405.0203 &F2&   389&   11.370&          &        &       &       2000    & 0.25 &\\
S10962&2405.1545 &B0e&  409&    9.716&          &  3580  & 0.12  &               &       &       see Fig.~\ref{S10962}  \\
S10963&2407.0022 &M3&   372&   11.554&          &        &       &       20$-$100  &  0.4   &        =NSV 2073\\
S10964&2408.0661 &B5&   397&    9.003&          &        &       &        3000   &  0.2    &       see Fig.~\ref{S10964} \\
S10965&2409.0265 &A2&   401&   11.091& ~$  -0.16$&        &       &
     &&\\ \hline 
\end{tabular}}
\parbox{16.8cm}{
$(1)$ Short-term variable, only enhanced scatter, no significant long-term variability.\\
$(2)$ New eclipsing binary, probably of EB type ($\beta$ Lyrae); epoch of minimum: HJD 24 37585.82. Possible alias 
    period:~$1\fd59315$.}
    %\end{center}
\end{table}